Author’s response to reviews

Title: Prevalence and predictors of irritable bowel syndrome in patients with morbid obesity: a cross-sectional study

Authors:

Martin Aasbrenn (Martin.Aasbrenn@gmail.com)
Ingvild Høgestøl (ingvild.hogestol@gmail.com)
Inger Eribe (inerib@ous-hf.no)
Jon Kristinsson (joakri@ous-hf.no)
Stian Lydersen (stian.lydersen@ntnu.no)
Tom Mala (tommal@ous-hf.no)
Per Farup (per.farup@ntnu.no)

Version: 2 Date: 23 May 2017

Author’s response to reviews:

Dear Editor,

We thank you for the new evaluation. As requested, some changes has been done in the document:

1. In the Declarations section we have added a heading “Acknowledgments” (page 17).

2. In the Declarations section we have added a heading “Consent to publish” (page 16).

3. The section “Keywords” is moved after the abstract (page 3).
4. The manuscript is uploaded as a single, final, clean version without any tracked changes or text in different colors.

We confirm that all authors have agreed to the authorship of this revised version of the manuscript.

Best wishes,

Martin Aasbrenn, MD

Martin.Aasbrenn@gmail.com